Author’s response to reviews

Title: Theoretical and perceived balance of power inside Spanish public hospitals

Authors:

Paloma Salvadores (p.salvadores@cs.urjc.es)
Jose Schneider (schneider@teleline.es)
Imanol Zubero (cipzubei@lg.ehu.es)

Version: 6 Date: 3 Sep 2001

Dear Dr Mongin-Bulewski,

enclosed you will find the final version of the manuscript, to which we have added all modifications kindly suggested by the referees, as listed below. In this respect, we would like to thank especially Prof. Charles Hoff for the trouble he has repeatedly taken in correcting our grammatical errors.

List of corrections

Prof. Charles Hoff:

"First": the grammatical corrections kindly suggested by him have been made (see above)
"Second": the labels of Fig. 1 have been changed
"Third": the Chi-square values have been included into the corresponding legends of the figures

Dr Groneberg

"The question of the coloured or non-coloured tables/charts and the length of the abstract should be discussed with the editor"

We leave these items entirely to the discretion of the editor.

Thank you very much once again for the incredibly swift handling of the manuscript.

With kindest regards,

Paloma Salvadores, Imanol Zubero, Jose Schneider